I would like to congratulate you on the work seriously done. Thank you for submitting your work to WJO. This is an important topic. I would like to make minor comments that were not very clear to me in the paper. Illustrations: The figures need to have the letters (Figure 1, a b c), despite being only of the manuscript. This is a case report, so I would suggest to remove "and literature review" from the title as no literature review was conducted and included. The language editing rating needs to be perfect, revise your article. Please, review and accept my corrections if they applicable.

- Thanks for the valuable suggestions from the reviewer!

- I added the letters to the images in the text

- I removed “and literature review”

- I revised my article